EPRU

Economic Policy Research Unit Institute of Economics University of Copenhagen Studiestræde 6 DK-1455 Copenhagen K DENMARK Tel: (+45) 3532 4411 Fax: (+45) 3532 4444 E-mail: Grethe.Mark@econ.ku.dk Homepage: http://www.econ.ku.dk/epru/

Sudden Stops and the Mexican Wave: Currency Crises, Capital Flow Reversals and Output Loss in Emerging Markets

Michael M. Hutchison and Ilan Noy

2002-12

ISSN 0908-7745

The activities of EPRU are financed by a grant from The Danish National Research Foundation

Sudden Stops and the Mexican Wave:

Currency Crises, Capital Flow Reversals and Output Loss in Emerging Markets

May 2002

Michael M. Hutchison Department of Economics Social Sciences 1 University of California, Santa Cruz Santa Cruz, CA 95064 Email: <u>hutch@cats.ucsc.edu</u> (Corresponding author) Ilan Noy Department of Economics Social Sciences 1 University of California, Santa Cruz Santa Cruz, CA 95064 Email: ilann@cats.ucsc.edu

*We thank participants at the UCSC workshop series on currency crises for helpful suggestions on an earlier version of this paper, especially Joshua Aizenman, Menzie Chinn and Ken Kletzer.

Abstract

Sudden Stops are the simultaneous occurrence of a currency/balance of payments crisis with a reversal in capital flows (Calvo, 1998). We investigate the output effects of financial crises in emerging markets, focusing on whether sudden-stop crises are a unique phenomenon and whether they entail an especially large and abrupt pattern of output collapse (a "Mexican wave"). Despite an emerging theoretical literature on Sudden Stops, empirical work to date has not precisely identified their occurrences nor measured their subsequent output effects in broad samples. Analysis of Sudden Stops may provide the key to understanding why some currency/balance of payments crises entail very large output losses, while others are frequently followed by expansions. Using a panel data set over the 1975-97 period and covering 24 emerging-market economies, we distinguish between the output effects of currency crises, capital inflow reversals, and sudden-stop crises. We find that sudden-stop crises have a large negative, but short-lived, impact on output growth over and above that found with currency crises. A currency crisis typically reduces output by about 2-3 percent, while a Sudden Stop reduces output by an additional 6-8 percent in the year of the crisis. The cumulative output loss of a Sudden Stop is even larger, around 13-15 percent over a three-year period. Our model estimates correspond closely to the output dynamics of the 'Mexican wave' (such as seen in Mexico in 1995, Turkey in 1994 and elsewhere), and out-of-sample predictions of the model explain the sudden (and seemingly unexpected) collapse in output associated with the 1997-98 Asian Crisis. The empirical results are robust to alternative model specifications, lag structures and using estimation procedures (IV and GMM) that correct for bias associated with simultaneity and estimation of dynamic panel models with country-specific effects.

JEL: F32, F43, O16

1. Introduction

"Sudden Stops" in capital flows to emerging market economies are a key characteristic of several recent financial crises. The sudden stop phenomenon, emphasized by Calvo (1998), features an abrupt cessation in foreign capital inflows and/or a sharp capital outflow concurrently with a currency/balance of payments crisis. Most currency/balance of payments crises are *not* characterized by Sudden Stops (Table 1). Capital inflow reversals occur with some regularity in emerging markets (about 22% of the observations in our sample), and currency crises are also fairly common (12% of the observations). But most of these occurrences are not the *joint* occurrence of capital flow reversals and currency/balance of payments crises-- Sudden Stops occur in only about 6% of the observations in our sample of emerging market economies.

Sudden Stops may have severe consequences for the economy, as the abrupt reversal in foreign credit inflows may cause a sharp drop in domestic investment, domestic production and employment. The adverse consequences of a sharp reversal in foreign capital inflows could be the reason that only a subset of currency/balance-of-payments crises in emerging market economies are found to be associated with deep recessions (Gupta et al., 2000). The pattern of a sudden cessation in capital inflows and currency crisis followed by an abrupt, but short lived, output collapse has been termed the "Mexican Wave" by the <u>Financial Times</u> in light of the Mexican experience in 1995. By our metric, there have been 24 episodes of Sudden Stops among emerging markets since the collapse of the Bretton Woods system of fixed exchange rate parities in the early 1970s (listed in the appendix).

Recent theoretical literature, following the work of Calvo (1998) and Calvo and Reinhart (2000), emphasizes the linkages between Sudden Stops and output losses (e.g. Aghion et al.,

2001; Mendoza, 2001; and Gopinath, 2001). Sudden Stops cause particularly large drops in output when economies face financial frictions associated, for example, with price stickiness combined with an external financing premium (e.g. Bernanke et al., 1999), debt-deflation combined with collateral constraints (e.g. Kiyotaki and Moore, 1997), and other forms of imperfect credit markets (e.g. Mendoza, 2001).

Empirical work to date has not systematically distinguished between different types of currency/balance of payments crises, and this may help to explain the mixed results of studies attempting to measure the output effects of financial crises. Ventura (2000), for example, notes the importance of distinguishing between different types of current account reversals, and by extension of currency crises, in attempting to evaluate their real output costs. Analysis of Sudden Stops may provide the key to understanding why some currency/balance of payments crises entail very large output losses, while others are frequently followed by expansions.

To address this issue, we investigate the output growth dynamics following currency crises, capital flow reversals and Sudden Stops in a panel data set of 24 emerging-market economies covering the 1975-97 period. We measure the impact of crises in a panel regression framework, carefully controlling for domestic and external factors, country time-invariant effects, and state of the business cycle. Simultaneity between financial crises and output growth is likely in this context, and we employ the fixed-effects panel IV and GMM estimation procedures, respectively, of Hausman and Taylor (1981) and Arellano and Bond (1991) to address this issue.

We find that sudden-stop crises have a large negative, but short-lived, impact on output growth over and above the effect found with currency crises. Our results also correspond very closely with the output dynamics of the 'Mexican wave' (such as seen in Mexico in 1995, Turkey

in 1994 and elsewhere), and out-of-sample predictions of the model explain the sudden (and seemingly unexpected) collapse in output associated with the 1997-98 Asian Crisis.

Section 2 reviews the literature on Sudden Stops and highlights our contribution. Section 3 presents the basic empirical model. Section 4 discusses the data employed in the study. Section 5 reports summary statistics on key macroeconomic variables and the primary empirical results of the study. This section presents estimation results of the output equations, model dynamics and robustness checks. Section 6 presents evidence as to the channel through which a sudden stop in capital inflows affects the real economy and how well the dynamics of the model correspond with the Mexican wave pattern of output losses. We also present predictions for output development in the East Asian crisis obtained by simulating our empirical results for the out-of-sample data for the five Asian 1998 crisis countries. Section 7 concludes the paper.

2. Why should a Sudden Stop cause a collapse in output?

The sudden stop phenomenon involves a reversal in capital inflows associated with a currency and balance of payments crisis. There are several reasons why one would expect a Sudden Stop to cause a severe recession. Calvo (1998, 2000) and Calvo and Reinhart (2000) analyse several mechanisms through which a sudden stop in international capital flows may bring about a currency and balance of payments crisis and the reasons that an output collapse may follow. The first mechanism may be termed the traditional Keynesian effect whereby a fall in credit, attributable to the sudden stop in capital inflows combined with an external financing premium and a "financial accelerator", reduces aggregate demand and causes a fall in output (e.g. Bernanke et al., 1999).

The second mechanism, termed the "Fisherian" channel by Mendoza (2001), emphasizes that a Sudden Stop enhances the severity of a currency crisis since it hits the financial sector and, given collateral constraints, induces a debt-deflation and a real contraction (e.g. Kiyotaki and Moore, 1997). Furthermore, firm bankruptcies may cause negative externalities-- banks may become more cautious and reduce loans. This is turn induces a further fall in credit—the "vanishing credit effect" described in Calvo (2000)--and contributes to recession. Credit that would normally automatically be rolled over is now conditioned upon passing more in-depth viability tests. The resulting "highway congestion" in credit markets amounts to a negative supply shock.

A number of new theoretical studies have developed models further formalizing the linkages between sudden stops in capital inflows, financial crises and declining output. This literature includes, among others, the works of Aghion, Baccheta and Banerjee (2000), Caballero and Krishnamurthy (2000 and 2001), Céspedes, Chang and Velasco (2001), Daniel (2002), Gopinath (2001), Mendoza (2001), and Schneider and Tornell (2001).

Empirical literature on Sudden Stops and output collapse

By contrast with the rapidly emerging theoretical literature linking Sudden Stops with output losses, there is no empirical literature formally addressing this issue. Several recent papers empirically analyze output developments around the time of currency crises in broad samples of countries (e.g. Aziz et al., 2000; Gupta et al., 2000; Hutchison and Noy, forthcoming). Similarly, Barro (2001), Bordo et al. (2001) and Hutchison and Noy (2002) measure output costs associated with the occurrence of both currency crises and banking crises.

No study of which we are aware, however, analyses Sudden Stops in the sense of a capital inflow reversal *in tandem with* a currency/balance of payments crisis. As Table 1 indicates, there are many instances of a currency crisis that are not associated with a capital flow reversal (about 50% of the cases). Our objective is to focus on the joint occurrences that we believe are the best empirical representations of the Sudden Stop problem.

Milesi-Ferretti and Razin (2000) and Edwards (2001) is the work most closely related to ours on Sudden Stops. Milesi-Ferretti and Razin (2000) analyse *separately* the output costs of current account reversals and the output costs of currency crises.¹ They consider output developments around instances of current account reversals (and, separately, currency crises), and analyse the factors that influence these developments in the context of a cross-sectional data set of developing and emerging market economies.² Unlike our paper, however, they do not attempt to measure directly the marginal effect of a crisis or reversal on output growth either separately or jointly (holding other macroeconomic and institutional factors constant) in a panel data set of emerging markets. Edwards (2001) considers the output effects of reversals, but not of financial crises.

Section 3. Estimating the Effects of Sudden-Stop Crises on Real Output Growth

Our contribution is to measure the output cost of a sudden-stop crisis. Unlike other literature that evaluates the outcomes of financial crises, we control for simultaneity issues, and biases associated with estimation of dynamic panel data models. Estimating this model for emerging market economies, we are also able to examine whether sudden-stop crises are unique phenomena in general terms and whether the deep recessions in East Asia were typical of the "bad" outcome associated with sudden-stop crises.

Our approach begins by explaining output growth in emerging markets by a standard set of variables. The determinants of output in this model are a set of domestic policy, structural, and external factors, as well as country-specific effects and lagged output growth. Domestic policy factors are changes in government budget surpluses and credit growth. External factors are growth in foreign output and real exchange rate overvaluation. The structural factor we consider is the openness of the economy to international trade. Country-specific fixed effects (in the context of panel data) are introduced in order to account for the widely varying growth experiences in our set of emerging-market economies over the past 25 years. All of the variables, with the exception of foreign output, are introduced with a one-year lag in order to capture the delayed response of output to macroeconomic developments. This formulation of the model avoids the potential for biased coefficient estimates on the domestic policy variables due to feedback effects from output growth to policy formulation (simultaneous equation bias).

Our main concern with the benchmark model is to introduce relevant control variables into the regression equation so that the identified impact of a crisis on output growth is not simply due to omitted-variables bias. We refrain from including a larger set of macroeconomic variables as these are typically not robust to model specification (Levine and Renelt, 1992).

In the context of our benchmark model, we test for the additional effect on output growth arising from a currency crisis (we define the currency crisis and capital account reversal variables in section 4). We consider both lagged and contemporaneous effects of crises on output growth, and also estimate several variants of the model, including changes in the lag structure and definition of crises, to check the robustness of the basic results. The coefficient estimates on our crisis measure may be interpreted as the marginal effects of crises, after controlling for several of the other factors that may influence the evolution of output growth. Additionally, we include a

'capital account reversal' variable and interpret its coefficient as the marginal effect of such a reversal on output growth. An interactive term of a capital account reversal and a currency crisis – which we label as a sudden-stop crisis – then permits us to directly estimate the effect of a sudden-stop crisis on the evolution of output.

The formal specification of the empirical model is as follows. The growth of real GDP for the *i*th country at time $t(y_{it})$ is explained by policy variables $(x_{i(t-1)})$; external and structural factors $(w_{i(\cdot)})$; the recent occurrence of a currency crisis and a current account reversal $(D_{i(\cdot)}^{CC}, D_{i(\cdot)}^{KA})$, a sudden-stop crisis $(D_{i(\cdot)}^{CC} * D_{i(\cdot)}^{KA})$, and an unobservable random disturbance (ε_{it}) .

$$y_{it} = \mu_0 + \beta_k x_{i(t-1)} + \alpha_h w_{i(\cdot)} + \beta^{CC} D_{i(\cdot)}^{CC} + \beta^{KA} D_{i(\cdot)}^{KA} + \beta^{SS} (D_{i(\cdot)}^{CC} \cdot D_{i(\cdot)}^{KA}) + \varepsilon_{it}$$
(1)

where x is a k-element vector of policy variables for country i at time t, w is an h-element vector of external variables for country i at times (t or t-1), $D_{i(t)}^{CC}$ is a dummy variable equal to unity if the country has recently experienced a currency crisis or balance of payments crisis (and zero otherwise) and likewise for a capital account reversal. The interactive Sudden-Stop term is therefore equal to unity if a currency crisis and a capital account reversal were observed in the same year. ε_{it} is a zero mean, fixed variance, disturbance term. μ_0 is a vector of country effects (allowing average growth rates to vary across countries in the sample), β_k is a k-element vector measuring the impact of policy changes on output, α_h is an h-element vector measuring the impact of exogenous factors on output, and β^{CC} , β^{KA} and β^{SS} measure the output growth effects of a currency crisis, a capital account reversal and a sudden-stop crisis, respectively. In our robustness regressions we examine possible non-linearities and size effects for the crises variables, a larger sample including developing countries and different estimation methods.

In our main estimates we follow a procedure first suggested by Hausman and Taylor (1981) that takes into account the bias in estimation of a dynamic panel with predetermined and endogenous variables (for a rigorous formulation of this bias, see Nickel, 1981). When a correlation exists between the independent variables and the individual country-specific effects, a least-squares estimation of a dynamic model ignores the correlation between the time-invariant country-fixed effects and the error term. A similar correlation between the "crisis" explanatory variables and the error term exists when output fluctuations contribute to the onset of a crisis. The Hausman-Taylor three-step estimation methodology is an instrumental variable and the individual country-specific effects, as well possible simultaneity issues running from output growth (our dependent variable) and currency crises and capital account reversals and their interaction (three of the explanatory variables).

In the first step, least squares estimates (with fixed effects) are employed to obtain consistent but inefficient estimates for the variance components for the coefficients of the time-varying variables. In the second step, an FGLS procedure is employed to obtain variances for the timeinvariant variables. The third step is a weighted IV estimation using deviation from means of lagged values of the time-varying variables as instruments.³ The procedure requires specifying which explanatory variables are to be treated as endogenous. In our specification, the endogenous explanatory variables are the three binary crisis measures (currency, reversals and Sudden Stops) and the lagged dependent variable.⁴

While the Hausman-Taylor (HT) procedure provides asymptotically unbiased estimates, a recent literature suggests it is not the most efficient estimator possible. A more efficient General Methods of Moments (GMM) procedure relies on utilizing more available moment conditions to obtain a more efficient estimation (e.g., Ahn and Schmidt, 1995; and Arellano and Bond, 1991 and 1998).⁵ This procedure, however, is usually employed in estimation of panels with a large number of individuals and short time-series such as in the literature on long-run growth (Bond et al., 2001). In our case, the data makes this procedure difficult to implement for most specifications of the model. We provide some results using the Arellano and Bond (1998) GMM framework. These coefficient estimates are very similar to those obtained from the benchmark Hausman-Taylor procedure.⁶

Section 4. Data Description

We concentrate our investigation on emerging markets since they are the focus of policy discussions and recent experiences of financial crises and output collapses. Several recent studies indicate that emerging markets may be different with respect to the factors that make them susceptible to a financial crisis (Glick and Hutchison, 2001; Caballero and Krishnamurthy, 2002) and how they respond to them (IMF, 1998). Specifically, emerging markets tend to be open to international capital inflows, and have experienced pre-crisis large private inflows that are typically short-term. Large short-term foreign-currency debt positions also increase the vulnerability of these economies to swings in exchange rates and cessation of new capital to roll over existing debt. Emerging markets therefore appear most vulnerable to sudden-stop crises and, potentially, their adverse consequences.

Defining Currency and Balance of Payments Crises

Our indicator of currency and balance of payments crises is constructed by identifying "large" values in an index of currency pressure, defined as a weighted average of monthly real exchange rate changes and monthly (percent) international reserve losses.⁷ Following convention (e.g. Kaminsky and Reinhart, 1999) the weights are inversely related to the variance of changes of each component over the sample for each country. This excludes some large depreciations that occur during high inflation episodes, but it avoids screening out sizeable depreciation events in more moderate inflation periods for countries that have occasionally experienced periods of hyperinflation and extreme devaluation.⁸ Our measure, taken from Glick and Hutchison (2001), presumes that any nominal currency changes or reserve changes associated with exchange rate pressure should affect the purchasing power of the domestic currency, i.e. result in a change in the real exchange rate (at least in the short run). An episode of serious exchange rate pressure, i.e. a standard crisis episode, is defined as a value in the index—a threshold point-- that exceeds the mean plus 2 times the country-specific standard deviation, provided that it also exceeds 5 percent.⁹ The first condition insures that, relative to its own history, unusually large values of the index of currency pressure are counted as a crisis while the second condition attempts to screen out values that are insufficiently large in an economic (real) sense.

For each country-year in our sample, we construct binary measures of currency crises, as defined above (1 = crisis, 0 = no crisis). A currency crisis is deemed to have occurred for a given year if the currency pressure index for any month of that year satisfies our criteria. To reduce the chances of capturing the continuation of the same currency crisis episode, we impose windows on our data. In particular, after identifying each indication of currency pressure, we treat any similar threshold point reached in the following 24-month window as a part of the same currency episode

and skip the years of that change before continuing the identification of new crises. With this methodology, we identify 51 currency crises, 68 crisis years and 40 major currency crises for our emerging markets dataset over the 1975-97 period (see appendix).

Defining Capital Account Reversals and Sudden-Stop Crises

As in Milesi-Ferretti and Razin (1998 and 2000) we identify capital account reversals by examining changes in the current account deficit. As our focus is on the short and medium term costs of sudden-stop events, we define a capital account reversal as a positive change in the annual current account surplus that is bigger than a pre-specified threshold (measured as a percentage of current GDP).¹⁰ Following Milesi-Ferretti and Razin (1998) we use thresholds of 3% (5%) for standard (major) current account reversals.¹¹ Similarly to our currency crisis variable, we construct, for each country-year, a binary measure of reversals, as defined above (1 = reversal, 0 = no reversal) for both thresholds and impose a 24-month window on our data.¹²

We use an alternative definition for capital account reversals in our robustness tests. We impose an additional condition that the post-reversal current account deficit is higher than -1% of GDP. This is motivated by the hypothesis that capital inflow reversals will be especially painful if they constitute an almost complete stop in capital inflows (or even capital outflows). In our robustness checks we also control for the size of the reversal (as a percent of GDP).

We define a sudden-stop crisis as one in which there is the contemporaneous occurrence of a currency crisis (a deviation in our currency pressure index of more than two standard deviations), and a capital account reversal (a change in the current account of more than 3% of GDP). As we examine contemporaneous occurrences, we do not speculate on the exact casualty structure between the two.

Control Variables in the Output Growth Equation

As discussed in section 2, the domestic policy factors included in our estimation are lagged changes in government budgets and lagged credit growth; external factors are (trade-weighted) external growth rates of the G-3 and lagged index of real exchange rate overvaluation; and the structural factor we consider is the openness of the economy to international trade.¹³ The macroeconomic data series are taken from the International Monetary Fund's IFS CD-ROM.

The minimum data requirements to be included in our study are that GDP figures are available for a minimum of 10 consecutive years over the period 1975-97. We use annual observations. We employ monthly data for our (real) exchange rate and international reserves pressure index to identify currency crises and date each by the year in which it occurs; monthly or quarterly data on current account deficits is available only for a sub-sample of our emerging markets sample so we use annual data for the identification of reversals.

Section 5. Empirical Results

Table 1 presents the number and frequency of currency crises, capital flow reversals and Sudden Stops for our data set. The top panel is for "normal" currency crises and the lower panel is for "major" currency crises. The panels are divided into four parts: the upper left quadrant shows the number and frequency of sudden stop crises (combination of a currency crisis and a capital flow reversal); the lower left quadrant reports the number and frequency of currency crises *not* associated with reversals; the upper right quadrant gives the number and frequency of capital flow reversals *not* associated with currency crises; and the lower right quadrant represent the number and frequency of tranquil episodes (neither a currency crisis nor a capital flow reversal).

Tranquil periods are the most common, of course, representing 72 percent of the sample observations (300 observations). Sudden-stop crises (24 observations) and currency crises (25 cases without capital reversals) each represent about 6 percent of the sample. Capital flow reversals without currency crises are more frequent, at about 16 percent of the sample. With the higher threshold levels for major crises shown in the lower panel, the frequency of occurrences is lower but the basic pattern is the same as with the standard crisis definitions. Clearly, suddenstop crises are *not* one and the same with currency crises nor with capital flow reversals. It is not uncommon to have capital flow reversals without currency crises and vice versa. Sudden Stops are distinct phenomena.

Conditional Probabilities for Crises Onsets

Table 2 presents hypothesis tests on the likelihood that currency crises and capital account reversals (both standard and major) are statistically independent. The hypothesis that they are not correlated when a currency crisis occurs contemporaneously with a capital account reversal *cannot* be rejected. Yet, the hypothesis that the occurrence of a capital account reversal is not associated with a currency crisis in the previous year can be clearly rejected with probability of more than 99%. Furthermore, there is no statistically significant correlation between contemporaneous currency crises and lagged reversals. While this observation is suggestive as to the possible causal links between the two, assuming a causal link is not necessary for the empirical work and discussion that follow.

Macro Developments: Before/After Crises Statistics

Table 3 presents summary statistics on key macroeconomic developments around currency crises (upper panel) and capital account reversals (lower panel). It presents before-after statistics for the standard definitions of a 'normal' currency crisis and the standard (3%) capital account reversal event as described in section 4. Four-year windows are imposed on the data to clearly delineate the macroeconomic developments around the time of crises.

Our focus variable, real GDP growth, shows an average decline of about 1.3 percentage points in the year a currency crisis takes place, and it recovers only minimally the following year (by 0.3 percentage points). Average output growth goes back to its previous level two years after the crisis, and this upturn is statistically significant. This pattern is almost identical for standard and major crises (not reported for brevity). Average losses appear to be somewhat bigger (smaller) for our sub-sample of currency crises with (without) a capital account reversal---reducing output growth by 1.5 (1.1) percentage points, respectively.

By contrast, output developments around capital account reversals are striking both by the larger costs involved (1.9 percentage points for the year of the reversal) and by the rapid growth turnaround. A capital account reversal is typically followed by an increase in output growth of 3.3 percentage points in the following year. Hence, at first pass, the summary statistics indicate significant and—in some cases—prolonged effects of financial crises and a short-lived but more pronounced effect of a capital account reversal.

Benchmark Model Estimates

Table 4 presents results from our benchmark model. The statistically significant control variables are external output growth, real exchange rate overvaluation, and lagged output growth.

A one- percent rise in the growth rate of the G-3 economies raises output growth in emergingmarket economies by about, on average, 0.3-0.4 percentage points. A rise in real exchange rate overvaluation significantly reduces output growth. This is noteworthy in its own right, indicating that emerging market economies should avoid currency overvaluation, but also because real exchange rate overvaluation is a reliable predictor of future currency crises (see Glick and Hutchison, 2001). However, the coefficients for budget changes, credit growth and the openness measure are not significantly different from zero. The coefficient estimates for the control variables are consistent across many alternative specifications of the model reported in columns (1)-(4) of table 4 and in the other tables.

Turning to the currency/balance of payments crisis variables, the coefficient estimates reported in column (2) indicate that the onset of a contemporaneous (lagged) currency crisis is associated with a fall in GDP growth of about 2.5 (2.6) percentage points. Very similar results are obtained, but not reported, when including only the contemporaneous or the lagged currency crisis binary variable. After a two-year period, the cumulative negative effect of a currency crisis on output is therefore about 5.1 percent.

Table 4 also presents more information on the dynamics of output adjustment to currency crises. Adding further lags (second, third and fourth year lags) to the model, reported in column (3), indicate that the contemporaneous and one-year ahead effects of a currency crisis remain negative and highly significant and with the same magnitudes as reported previously. This is followed by statistically insignificant effects on the second, third and fourth year following a crisis. This result remains when some of the insignificant lags are dropped. Our results therefore indicate that the output costs of a currency crisis do not extend beyond a two-year horizon.

We also include lead values of currency crises in the equations, shown in column (4), to further investigate the dynamic responses. The one-year lead coefficient for the currency crisis variable is statistically significant. This result indicates that a currency crisis tends to follow a decline in real output growth. On the other hand, a currency crisis is also associated with a further decline in output growth contemporaneously and over a period of two years. These model estimates suggest that, within 4 years, output declines cumulatively by almost 8 percent for an average currency crisis in an emerging-market economy.

An important question is whether a particularly severe crisis—substantially larger than the normal crisis—has an especially severe effect on growth. To investigate this issue, we introduce a "major" currency crisis variable that is identified by a threshold point in our pressure index that exceeds 3-standard deviations from the mean. For brevity we do not report these results. Somewhat surprisingly, the output effects of a major crisis are not larger than the typical crisis situation. Coefficients for a version of column (2) using the major crisis measure yield coefficients of -2.3 and -2.8 for the contemporaneous and lagged major currency crisis variables, respectively. Major currency and balance of payments crises therefore do not appear to have a substantially different impact on output growth than the average crisis (identified using a 2-standard deviation threshold).

Reversals and Sudden Stops

The full results for our model are reported in Table 5. Columns (1) and (2) report the cost of a capital account reversal for the 5 percent and 3 percent thresholds, respectively, with the inclusion of lagged and contemporaneous currency crises variables. In both cases, reversals are costly. Surprisingly, "normal-sized" reversals (3 percent of GDP) seem to have, on average, a

larger effect on output growth (with almost 2 percentage points decline) than do large reversals (1.2 percentage point decline).¹⁴

Our main results are presented in columns (3)-(4). The coefficient on the sudden-stop crisis interactive variable is negative, *large* and statistically significant. The output costs of sudden-stop crises appear to be very large—a drop in output growth of 4.7-6.5 percentage points in the same year of the crisis. Furthermore, while the coefficients on the currency crisis variables are smaller they are still statistically significant for the 99% confidence level. Neither does the inclusion of leads and lags for the capital account reversal dummy, reported in column (5), change the magnitude of these coefficients.¹⁵ We find that a sudden-stop crisis has a very large average effect on output growth—depressing GDP by about 10 percent over a short (2-year) period.¹⁶

Robustness Tests

To check the robustness of our results we first examine whether our estimation technique, based on the Hausman and Taylor (1981) IV estimator, gives similar coefficient estimates to those obtained by the least-squares fixed effects estimator with a White heteroscedasticity correction (LSDV) or the more efficient first-differenced GMM estimator suggested by Arellano and Bond (1991 and 1998). These results are reported in Table 6, columns (1)-(3), where we also include the HT estimation for exactly the same sample.¹⁷ There is relatively little difference between the coefficients obtained on our focus variables—currency crises and reversals—in all three estimation techniques. As can be expected, the GMM estimator yields higher t-statistics.

We also run the same model for a larger sample including 42 developing countries as well as the emerging markets sample. Data availability guided our choice of additional countries.¹⁸ A

comparison of column (4) with column (2) in Table 5 leads us to conclude that both currency crises, reversals and Sudden Stops have a weaker impact on output growth in our larger sample of developing countries-- -1.5 instead of -1.9 for lagged currency crises, -1.1 instead of -2.0 for capital account reversals and -3.6 instead of -6.6 for sudden-stop crises. Column (5) adds a lagged currency crises variable, which further reduces the coefficient on our sudden-stop variable. The larger set of developing countries does indeed seem to be less vulnerable, on average, to both turmoil in currency markets and reversals in capital flows than emerging markets. This may simply be attributable to the fact that poor developing countries attract very little private capital inflows and hence a substantial reversal observable in the current account is different.

In column (6) of Table 6 we investigate whether the main results are robust when we account for the size of the capital account reversal. To our central regression--reported in table 5 column (4)--we add the size of the capital flow reversal as a percentage of GDP. The coefficient on the reversal's size variable is statistically different from zero and indicates that any flow reversal of one percent (of GDP) reduced output growth by 0.2 percentage points. More central to our argument is the finding that the coefficient on the sudden-stop dummy does not change much—indicating that non-linearities are important in understanding the effects of crises. A Sudden Stop is found to constitute a unique event that is important above and beyond the actual size of the reversal.

In column (7) of table 6 we modify the definition used for a current account reversal and its corresponding sudden-stop interactive term. We now use only a subset of our reversal observations—only those for which the reversal actually meant a full stop of capital inflows. Technically, we discard reversal observations for which the current account deficit, following the

reversal event was still bigger than 1 percent of GDP. As can be expected the coefficient on the newly defined sudden-stop variable is now even bigger and indicates a drop of 8 percentage points in GDP growth in the same year of the crisis. This result further supports our other findings on the very substantial negative average effects of Sudden Stops.

It is possible that the results reported to this point are subject to selection bias. Countries that experience a currency crisis may be different in important respects from other countries or episodes. That is, it may not be the currency crisis per se but several other factors contributing to them that are causing the decline in output growth. This is a variant of the selection bias problem.

We employ Heckman's (1979) Inverse Mills Ratio (IMR) to control for selection bias of this form. This statistic is constructed from the results of a probit regression explaining currency crises and added as an additional explanatory variable in the output growth regressions.¹⁹ Including the IMR in the regression of interest prevents possible bias in our coefficient estimates and is a standard approach.²⁰ For brevity, these results are not reported. In no case is the IMR coefficient statistically significant and, assuming the probit equation was correctly specified, selection bias may be rejected. More importantly, the coefficient estimates on the other explanatory factors, both the control and the crisis variables, are very similar to those reported in table 5.

Section 6. Predictive Accuracy, Dynamics and Channels of Transmission

Out of Sample Predictions

The models in the previous section explain up to 45 percent of the variation in output growth in our sample of emerging market economies. The sudden-stop proxy variables are highly significant and contribute to the overall explanatory power of the models. As an additional test of the robustness of the findings, we consider out-of-sample predictions for countries involved in the 1997 Asian crisis and subsequent collapse in output. All of these countries experienced Sudden Stops. In particular, Table 7 presents the predicted values for output growth for the five East Asian countries that experienced a severe financial crisis in 1997 and large output contractions in 1998—Indonesia, Korea, Malaysia, the Philippines and Thailand. These predictions are for 1998 output growth rates and are based on 1997 values of the explanatory variables and the coefficient estimates obtained from the model presented in column (4) of Table 5. Predicted values are decomposed into three different groups of explanatory variables: (a) domestic factors (lagged output growth, change in budget surplus, credit growth, and countryspecific effects); (b) external/structural factors (external growth, real exchange rate overvaluation and openness); and (c) the currency crisis, the capital account reversal and the sudden-stop crises.

Predicted output growth (from the Sudden Stop model) for all five countries is around negative 5 percent in 1998. Output is predicted to be worst for the Philippines (-5.2%) and best for Korea (-4.5%), with intermediate predictions for Indonesia, Malaysia and Thailand. The forecast errors (unexpected declines in output) are therefore substantial as the growth performances of the Philippines and Korea were radically different in 1998. At the extremes, the Philippines experienced the mildest recession (-0.5 output growth rate) while Indonesia experienced a -14.1 percent contraction of its economy. More importantly for our focus, the contemporaneous negative effect of a sudden-stop crisis is dominating our predictions with a cumulative negative effect of -9.4 percentage point decline in GDP growth. There is a positive domestic effect – mainly a history of very strong growth in the region and the consequently large country-specific effects-- and a modestly supportive external structural growth environment.

Although the forecast errors of the sudden-stop model are fairly large, the average prediction of a substantial decline in output (-5.0 percent) across four of the East Asian countries (Korea, Malaysia, Philippines and Thailand) is quite close to the actual average decline in output (-6.5 percent). Indonesia, perhaps due to political and social turmoil, seems to be a special case and experienced a –14.2 percent drop in output.²¹ Moreover, besides the sudden-stop model, we are not aware of other models that forecast (out-of-sample) large *declines* in output for East Asia in 1998. Hutchison and Noy (2002), for example, examined the output effects of currency, banking and "twin" crises. They conclude that "twin" crises lead to significant real costs, but cannot explain the depth of the recessions and the rapid recovery that the Asian-5 experienced between 1997 and 1999.

Dynamics and the Mexican Wave

The V-shaped output developments of the Asian-5 following the financial crisis also appear to fit the pattern associated with the Mexican Wave. Panel A of Figure 1 presents the output developments around some of the most famous sudden-stop crises: Turkey (1994), Mexico (1995), Indonesia (1998) and Korea (1998). Turkey, Indonesia and Korea follow the same output dynamics as Mexico--sharp output declines at the time of the crisis followed by large rebounds.

The sudden-stop model predictions, shown in Panel B of Figure 1, also correspond closely with the country experiences shown above. The panel shows the dynamic predictions for output around the time of a (i) currency crisis, (ii) a capital flow reversal, and (iii) a Sudden Stop.²² Both currency crises and capital flow reversals, taken alone, are associated with modest declines in output. Taken together, however, they constitute a Sudden Stop and induce a very large but short-lived drop in output. The model estimates mimic very closely the actual dynamics of Sudden Stops observed in Mexico, Turkey, Indonesia, and Korea.

Channels of Transmission: Sudden Stops and Collapsing Investment

The sudden-stop theory reviewed in section 2 discussed several potential transmission mechanisms through which Sudden Stops could cause an output collapse. A common element in these explanations is that a Sudden Stop causes domestic investment to collapse, perhaps through financing constraints that sharply limit imported investment goods or imported intermediate goods.

This view appears to be borne out. Table 8 shows average domestic investment growth, domestic fixed investment growth, export growth, and import growth around the 24 sudden-stop crises that we have identified in our emerging-market sample. Investment and imports are sharply reduced at the time of Sudden Stops. In particular, investment (import) growth was 8.5% (11.0%) two years prior to the Sudden Stop, decreases to 0.8% (2.0%) one-year prior, and then shows a large 6.0% (2.5%) *drop* at the time of the Sudden Stop. The declines in investment and imports continue for two years, and then sharply rebound two years after the sudden-stop crisis. By contrast, exports boom both the year of the crisis and in the year following the Sudden Stop.

This interpretation is consistent with the findings of Edwards (2001). Estimating investment equations for a large number of countries (industrial, developing and emerging) over 1970-97, he finds that current account reversals significantly reduce investment (relative to GDP). A 3 percent or larger reversal in the current account as a fraction of GDP (from deficit to surplus) reduces investment by about 3 percent of GDP within a two period.

These stylised facts are consistent with several case studies. Agénor et al. (2000) find, for example, that the sharp contraction in bank lending accompanying Thailand's financial crisis was due to a "supply crunch" on credit (presumably due to foreign financial constraints). Ghosh and Ghosh (1999) report similar findings for Korea and Indonesia as do Ito and da Silva (1999). Microeconomic research into firms' responses to changes in the exchange rate also suggests that firms generally reduce their investment when facing large domestic currency depreciation (Forbes, 2001 and Nucci and Pozzolo, 2001).

Section 7. Conclusions

The sudden-stop phenomenon seems to provide the key to understanding why some currency/balance of payments crisis entail very large output losses, while others are frequently followed by expansions. Using a panel data set over the 1975-97 period and covering 24 emerging-market economies, we distinguish between the output effects of currency crises, capital inflow reversals, and sudden-stop crises. We find that sudden-stop crises have a large negative, but short-lived, impact on output growth; and that these effects are substantially larger (almost three times greater) than those associated with a currency crisis alone. The empirical results are robust to alternative model specifications, lag structures and using estimation procedures (IV and GMM) that correct for bias associated with simultaneity and estimation of dynamic panel models with country-specific effects.

The sudden-stop model predictions also mimic the output dynamics of many countries facing sudden-stop crises. The Mexican Wave pattern that the model predicts was seen not only in Mexico at the time of the 1995 crisis, but also in such disparate countries facing Sudden Stops as Turkey, Indonesia and Korea. The channel of transmission also appears consistent with theory that points to an external "credit crunch" as a key element in capital flow reversals at the time of a currency crisis. Sudden Stops are associated with a collapse of imported goods and a dramatic fall in domestic investment. Clearly, the large output costs associated with Sudden Stops are a policy concern; even more so if the underlying cause of the capital inflow reversal and currency crisis are not attributable to "fundamentals" but rather to multiple equilibria or imperfections in the working of international capital markets.

References

Agénor, Pierre-Richard, Joshua Aizenman and A. Hoffmaister (2000). "The Credit Crunch in East Asia: What Can Bank Excess Liquid Assets Tell Us?" NBER Working Paper 7951, October.

Aghion, Philippe, Philippe Bacchetta and Abhijit Banerjee (2001). "Currency Crises and Monetary Policy in an Economy with Credit Constraints". *European Economic Review*, 45, 1121-1150.

Ahn, Seung C. and Peter Schmidt (1995). "Efficient Estimation of Models for Dynamic Panel Data," *Journal of Econometrics*, 68(1), 5-27.

Arellano, Manuel and Stephen Bond. (1991). "Some tests of specification for panel data: Monte Carlo evidence and an application to employment equations," *Review of Economic Studies*, 58, 277-297.

Arellano, Manuel and Stephen Bond (1998). "Dynamic Panel Data Estimation Using DPD98 for Gauss," Manuscript. Available: http://www.cemfi.es/~arellano/#dpd.

Arellano, Manuel and Bo Honoré (forthcoming). "Panel Data Models: Some Recent Developments," in J. Heckman and E. Leamer (eds.) *Handbook of Econometrics*. Amsterdam, New York and Oxford: Elsevier Science, North-Holland.

Aziz, Jahangir, Francesco Caramazza and Ranil Salgado (2000). "Currency Crises: In Search of Common Elements," IMF Working Paper WP/00/67 (March).

Bagnai, Alberto and Stefano Manzocchi (1999). "Current Account Reversals in Developing Countries: The Role of Fundamentals". *Open Economies Review*, 10, 143–163.

Barro, Robert (2001). "Economic Growth in East Asia Before and After the Financial Crisis," NBER Working Paper 8330 (June).

Bernanke, Ben, Mark Gertler and Simon Gilchrist (1999). "The Financial Accelerator in a Quantitative Business Cycle Framework". In Taylor, John and Michael Woodford (eds.) *Handbook of Macroeconomics*. Amsterdam, New York and Oxford: Elsevier Science, North-Holland.

Blundell, Richard and Monica Costa Días (2000). "Evaluation Methods for Non Experimental Data," *Fiscal Studies*, 21(4), 427-468.

Bond, Stephen, Anke Hoeffler and Jonathan Temple (2001). "GMM Estimation of Empirical Growth Models". Manuscript (September).

Bordo, Michael, Barry Eichengreen, Daniela Klingebiel and Maria Soledad Martinez-Peria (2001). "Is the Crisis Problem Growing More Severe?" *Economic Policy*, 16(32), April, 53-82.

Caballero, Ricardo and Arvind Krishnamurthy (2000). "Emerging Market Crises: An Asset Markets Perspective". Manuscript (February).

Caballero, Ricardo and Arvind Krishnamurthy (2001). "Smoothing Sudden Stops". NBER Working Paper 8427, (August).

Caballero, Ricardo and Arvind Krishnamurthy (2002). "A Dual Liquidity Model for Emerging Markets". NBER Working Paper 8758, (January).

Calvo, Guillermo (1998). "Capital Flows and Capital-Market Crises: The simple economics of Sudden Stops," *Journal of Applied Economics*, 1(1), November, 35-54.

Calvo, Guillermo (2000). "Balance-of-Payments Crises in Emerging Markets: Large Capital Inflows and Sovereign Governments" in Krugman, Paul (ed.) *Currency Crises*, Chicago, Illinois: University of Chicago Press.

Calvo, Guillermo and Carmen Reinhart (2000). "When Capital Inflows Come to a Sudden Stop: Consequences and Policy Options," in Peter Kenen and Alexander Swoboda (eds.) Reforming the International Monetary and Financial System, Washington D.C., The International Monetary Fund.

Céspedes, Luis P., Roberto Chang and Andrés Velasco (2001). "Balance Sheets and Exchange Rate Policy". Manuscript (December).

Daniel, Betty C. (2002). "Real Shocks and Financial Crises in Emerging Markets". Manuscript, April.

Domaç, Ilker and Giovanni Ferri (1999). "The Credit Crunch in East Asia: Evidence From Field Findings on Bank Behavior and Policy Issues". Manuscript, The World Bank (November).

Edwards, Sebastian (2001). "Does the Current Account Matter?" NBER Working Paper 8275, (May).

Forbes, Kristin (2001). "Cheap Labor Meets Costly Capital: The Impact of devaluations on Commodity Firms". Manuscript, August.

Frankel, Jeffrey and Andrew Rose (1996). "Currency Crashes in Emerging Markets. An Empirical Treatment," *Journal of International Economics*, 41, November, 351-366.

Ghosh, Swati R. and Atish R. Ghosh (1999). "East Asia in the Aftermath: Was There a Crunch?" Manuscript, The World Bank.

Glick, Reuven and Michael Hutchison (2000). "Capital Controls and Exchange Rate Instability in Developing Economies," Federal Reserve Bank of San Francisco *Center for Pacific Basin Studies Working Paper* No. PB00-05 (December).

Glick, Reuven and Michael Hutchison (2001). "Banking and Currency Crises: How Common Are Twins?" In R. Glick, R. Moreno, and M. Spiegel (eds.) *Financial Crises in Emerging Markets*. Cambridge, UK: Cambridge University Press.

Gopinath, Gita (2001). "Lending Booms, Sharp Reversals and Real Exchange Rate Dynamics". Manuscript, University of Chicago, GSB (November).

Gourinchas, Pierre-Olivier, Rodrigo Valdés and Oscar Landerretche (2001). "Lending Booms: Latin America and the World". NBER Working Paper 8249 (April).

Greene, William (2001). "Manual for LimDep" manuscript.

Gupta, Poonam, Deepak Mishra and Ratna Sahay (2000). "Output Response During Currency Crises," Manuscript, IMF (May).

Hausman, Jerry and William Taylor (1981). "Panel Data and Unobservable Individual Effects," *Econometrica*, 49(6), November, 1377-1398.

Heckman, James (1979). "Sample Selection Bias as a Specification Error," *Econometrica* 47 January, pp. 153-61.

Heckman, James (1990). "Varieties of Selection Bias," *American Economic Review: Papers and Proceedings* 80(2) May, pp. 313-318.

Hutchison, Michael and Ilan Noy (forthcoming). "Output Costs of Currency and Balance of Payments Crises in Emerging Markets," *Comparative Economic Studies*.

Hutchison, Michael and Ilan Noy (2002). "How Bad are Twins? Output Costs of Currency and Banking Crises". *UC Santa Cruz, Santa Cruz Center for International Economics Working Paper* 02-1, January.

International Monetary Fund (1998). "Financial Crises: Characteristics and Indicators of Vulnerability," *World Economic Outlook*, May.

Ito, Takatoshi and Luiz Pereira da Silva (1999). "New Evidence of Credit Crunch in Thailand and Indonesia and its Policy Implications". Manuscript, the World Bank.

Kaminsky, Graciela and Carmen Reinhart (1999). "The Twin Crises. The Causes of Banking and Balance-of-Payments Problems," *American Economic Review*, 89(3), June, 473-500.

Kiyotaki, Nobuhiro and John Moore (1997). "Credit Cycles," *Journal of Political Economy* 105(2), April, 211-248.

Levine, Ross and David Renelt (1992). "A Sensitivity Analysis of Cross-Country Growth Regressions". *American Economic Review*, 82(4), September, 942-963.

Mendoza, Enrique G. (2001). "Credit, Prices, and Crashes: Business Cycles with a Sudden Stop". NBER Working Paper 8338 (June).

Milesi-Ferretti, Gian Maria and Assaf Razin (1998). "Sharp Reductions in current account deficits: An Empirical Analysis," *European Economic Review* 42, 897-908.

Milesi-Ferretti, Gian Maria and Assaf Razin (2000). "Current Account Reversals and Currency Crises: Empirical Regularities," in Krugman, Paul (ed.) *Currency Crises*, Chicago, Illinois: University of Chicago Press.

Nickell, Stephen (1981). "Biases in Dynamic Models with Fixed Effects," *Econometrica*, 49(6), November, 1417-1426.

Nucci, Francesco and Alberto F. Pozzolo (2001). "Investment and the Exchange Rate: An Analysis with Firm-Level Panel Data". *European Economic Review*, 45, 259-283.

Rodrik, Dani (1999). "Where Did All The Growth Go? External Shocks, Social Conflict, and Growth Collapses." *Journal of Economic Growth*, 4, 358-412 (December).

Schneider, Martin and Aaron Tornell (2001). "Boom-Bust Cycles and the Balance Sheet Effect". Manuscript, UCLA (December).

Ventura, Jaume (2000). "Comment on Milesi-Ferretti and Razin" in Krugman, Paul (ed.) *Currency Crises*, Chicago, Illinois: University of Chicago Press.

Table 1 – Sudden Stops ^aNumber of Events (Percentage of Sample)

'Normal' Crises and Capital Flow Reversals					
	Currency Crisis	No Crisis			
Capital Flow	Sudden-Stop Crisis	65 (169/.)			
Reversal	24 (6%)	03 (10 /0)			
No Dovorcol	25 (60/)	Tranquil period			
no neversar	25 (0 %)	300 (72%)			
Major C	rises and Major Capital]	Flow Reversals			
	Currency Crisis	No Crisis			
Capital Flow	Sudden-Stop Crisis	35 (0%)			
Reversal	18 (4%)	55 (770)			
No Roversel	22 (5%)	Tranquil period			
	<i>44</i> (370)	339 (82%)			

^a For currency crises, a standard crisis is defined as a deviation of the currency pressure index of more than 2 standard deviations from the country-specific mean (3 standard deviations for major crises). For the current account reversal, a standard reversal is defined as a change in the current account to GDP ratio of more than 3 percentage points (5% points for a major CA reversal). The number in parentheses is the percent out of country years in our sample.

	3%	5%
	threshold	threshold
% of CA reversals associated with a	10	14
contemporaneous onset of currency	(0.88)	(0.26)
% of currency crisis associated with a	20	18
contemporaneous CA reversal	(0.88)	(0.26)
% of CA reversals associated with a	20***	23***
previous (t-1) currency crisis	(0.00)	(0.00)
% of currency crisis associated with a	46 ***b	31***
following (t+1) CA reversal	(0.00)	(0.00)

Table 2 - Conditional Probabilities ^a

^a chi square probability of independence of the two series in parenthesis.

Table 3 - Descriptive Statistics ^a

A. Currency Crises							
Dool CDD growth	Currency	4.0	2.7	1.4	1.7	3.6*	
rato (%)	Currency (no reversal)	4.2	3.5	2.4	2.2	3.4	
	Sudden Stop	3.8	2.0	0.5	1.3	3.8*	
Change in budget	Currency	-1.2	-1.4	-1.1	-1.0	-1.1	
surplus (%)	Currency (no reversal)	-1.8	-1.7	-0.9	-1.5	-1.2	
	Sudden Stop	-0.8	-1.1	-1.2	-0.6	-1.2	
Inflation rate (%)	Currency	30.5	31.6	36.3	43.4	42.0	
	Currency (no reversal)	36.4	38.7	36.2	40.5	47.3	
	Sudden Stop	23.3	23.0	34.4	44.0	34.9	
Our man the second to	Currency	-4.4	-4.3	-4.0	-1.4*	-0.4	
GDP ratio (%)	Currency (no reversal)	-2.8	-2.5	-3.0	-2.1	0.9*	
	Sudden Stop	-5.7	-4.8	-0.8*	-1.3	-2.2	
B. Capital Account Reversals							
Real GDP growth ra	5.4	4.6	2.7**	6.0***	5.9		
Change in budget s	urplus (%)	-0.9	-0.5	0.1	-0.6	0.1	
Inflation rate (%)		18.2	19.5	30.9	17.4	12.6	
Current account to GI	DP ratio (%)	-5.5	-7.5	0.9***	0.8	-0.9	

^a *, **, and *** denote rejection of same mean as the number to the left with 10, 5 and 1 percent confidence levels.

Dependent Variable: real GDP growth rate (DLRGDP)

Real GDP growth (t-1)	0.327***	0.266***	0.249***	0.240***
	(6.75)	(5.47)	(4.87)	(4.72)
Change in budget surplus to real GDP ratio (t-1)	-1.601	-0.385	-0.678	-2.460
	(-0.21)	(-0.05)	(-0.09)	(-0.32)
Credit growth (t-1)	-0.009	-0.007	-0.007	-0.006
	(-1.38)	(-1.05)	(-1.04)	(-0.88)
External growth rates - weighted average (t)	0.360***	0.381***	0.386***	0.390***
	(3.42)	(3.75)	(3.77)	(3.81)
Real exchange rate overvaluation (t-1)	-0.024*	-0.028**	-0.029**	-0.019
	(-1.88)	(-2.28)	(-2.28)	(-1.39)
Openness (t)	0.011	0.010	0.010	0.010
	(1.05)	(0.97)	(0.95)	(0.89)
Currency crises onset dummy - lead (t+2)				-0.457
				(-0.66)
Currency crises onset dummy - lead (t+1)				-1.558**
				(-2.21)
Currency crises onset dummy (t)		-2.453***	-2.622***	-2.793***
		(-3.78)	(-3.93)	(-4.16)
Currency crises onset dummy - lag (t-1)		-2.634***	-2.671***	-2.838***
		(-3.77)	(-3.64)	(-3.87)
Currency crises onset dummy - lag (t-2)			-1.008	-1.034
		-	(-1.31)	(-1.35)
Currency crises onset dummy - lag (t-3)			0.545	0.470
			(0.71)	(0.61)
Currency crises onset dummy - lag (t-4)			-0.066	0.010
			(-0.08)	(0.01)
Adjusted R ² ^e	0.27	0.32	0.32	0.33
Number of observations	374	373	370	370
Correlation of error terms	0.09	0.09	0.11	0.12

^a The Adjusted R² reported is for the fixed-effects least squares stage in the Hausman-Taylor procedure.

Table 5 - Growth Equation – HT - Current Account Reversals
Dependent Variable: real GDP growth rate (DLRGDP)

Real GDP growth (t-1)	0.218***	0.207***	0.207***	0.239***	0.197***
	(4.46)	(4.31)	(4.32)	(5.04)	(4.02)
Change in budget surplus to GDP ratio (t-1)	1.212	2.351	3.607	4.146	11.706
	(0.16)	(0.33)	(0.51)	(0.58)	(1.61)
Credit growth (t-1)	-0.009	-0.009	-0.007	-0.007	0.002
	(-1.47)	(-1.49)	(-1.16)	(-1.25)	(0.26)
External growth rates - weighted average	0.399***	0.394***	0.393***	0.390***	0.341***
	(4.03)	(4.05)	(4.09)	(4.01)	(3.36)
Real exchange rate overvaluation (t-1)	-0.033***	-0.034***	-0.036***	-0.037***	-0.037***
	(-2.70)	(-2.83)	(-3.11)	(-3.14)	(-3.19)
Openness	0.019*	0.018**	0.010	0.011	0.014
	(1.91)	(2.05)	(1.11)	(1.20)	(1.15)
Current Account Reversal (5% threshold)	-1.122*				
	(-1.75)				
Lead for Current Account Reversal (3%) (t+1)					-0.683
					(-0.90)
Current Account Reversal (3% threshold)		-1.958***	-1.243**	-0.924*	-1.685***
		(-3.88)	(-2.33)	(-1.75)	(-3.03)
Lag for Current Account Reversal (3%) (t-1)					-0.480
					(-0.58)
currency crises dummy (t)	-2.862***	-2.961***	-1.902***		-2.029***
	(-4.92)	(-5.18)	(-2.97)		(-3.09)
currency crises dummy (t-1)	-2.333***	-2.052***	-1.894***	-1.874***	-1.425**
	(-3.73)	(-3.33)	(-3.11)	(-3.04)	(-2.33)
sudden-stop crises dummy (t)			-4.727***	-6.593***	-5.066***
			(-3.59)	(-5.63)	(-3.80)
Adjusted R ²	0.35	0.37	0.39	0.38	0.44
Number of observations	374	374	374	374	320
Correlation of error terms	0.10	0.08	0.08	0.07	0.19

	(1) LSDV	(2) HT	(3) GMM1	(4) HTª	(5) HTª	(6) HT	(7) HT⁵
Real GDP growth (t-1)	0.133**	0.175***	0.193***	0.262***	0.257***	0.228***	0.263***
	(2.23)	(3.03)	(7.74)	(6.81)	(6.73)	(4.73)	(5.31)
Change in budget surplus to GDP ratio (t-1)	-12.296	-9.492	5.972**	7.191*	6.420	13.751*	8.987
	(-1.30)	(-1.05)	(2.01)	(1.66)	(1.49)	(1.85)	(1.19)
Credit growth (t-1)	-0.007	-0.008	0.003*	-0.001	-0.002	0.001	-0.002
	(-0.83)	(-1.09)	(1.96)	(-0.36)	(-0.61)	(0.16)	(-0.23)
External growth rates (t)	0.394***	0.387***	0.360***	0.339***	0.343***	0.326***	0.308***
	(3.31)	(3.29)	(8.75)	(4.29)	(4.38)	(3.19)	(2.95)
Real exchange rate overvaluation (t-1)	-0.041**	-0.045***	-0.033***	-0.026***	-0.025***	-0.038***	-0.037***
	(-2.46)	(-2.82)	(-6.40)	(-3.26)	(-3.11)	(-3.25)	(-3.07)
Openness	0.053*	0.021*	0.016***	0.016*	0.023**	0.025**	0.023*
	(1.67)	(1.88)	(8.98)	(1.84)	(2.53)	(2.00)	(1.83)
Current Account Reversal (3%) (t)	-1.972***	-2.050***	-1.249***	-1.075**	-1.355***	0.169	0.421
	(-3.28)	(-3.45)	(-11.90)	(-2.40)	(-3.01)	(0.17)	(0.55)
Currency crises dummy (t)					-1.624***		
					(-3.31)		
Currency crises dummy (t-1)	-2.612***	-2.492***	-2.951***	-1.515***	-1.421***	-1.328**	-1.892***
	(-3.29)	(-3.19)	(-6.24)	(-3.26)	(-3.09)	(-2.15)	(-3.02)
sudden-stop crises dummy (t)				-3.550***	-1.957*	-6.554***	-8.021***
				(-3.90)	(-1.92)	(-5.40)	(-5.76)
Reversal's size (t)						-0.234*	
						(-1.72)	
Sample	280	280	280	550	550	320	320

^a The sample contains also developing countries (with per capita income of less the 2000 in PPP\$ for 1992.

 $^{\circ}$ C(7) includes a different definition of current account reversals. In addition to (1) a change of at least 3% of GDP, it includes (2) the post reversal current account is above -1% of GDP.

	Domestic variables ^b	External variables	Currency crises	CA reversal	Sudden stop	Predicted growth	Actual growth
Indonesia	3.69	0.58	-1.87	-0.92	-6.59	-5.11	-14.16
Korea	3.78	1.13	-1.87	-0.92	-6.59	-4.47	-6.92
Malaysia	3.83	0.54	-1.87	-0.92	-6.59	-5.01	-7.65
Philippines	3.74	0.40	-1.87	-0.92	-6.59	-5.24	-0.54
Thailand	3.63	0.81	-1.87	-0.92	-6.59	-4.94	-10.73

Table 7- Out of Sample Growth Forecasts for East Asian Crisis Countries – 1998 ^a

^a Estimates are based on Table 5 column (4) coefficients. ^b Also includes the time invariant (fixed) effects.

	t-2	t-1	t	t+1	t+2
Domestic investment growth	8.50	0.84	-5.96	-4.81	8.07
Domestic fixed investment growth	6.87	2.81	-4.53	-7.82	3.83
Export growth	5.41	2.49	13.05	9.96	3.31
Import growth	11.04	2.03	-2.55	-1.03	11.51
Exports ^b	17.98	17.97	20.17	22.76	24.41
Imports ^b	21.00	21.96	20.91	19.20	21.87

Table 8 – Investment, Exports and Imports Around Sudden-Stops ^a

^a Data is from the World Bank's *World Development Indicators, 2001.* ^b Average in billion constant 1995 US\$.





Panel A – Recent Sudden Stops

Panel B – Sudden Stop Model Estimation



Appendix

Data for Currency and Sudden-Stop Crises

	Currency Crises	Sudden-Stop Crises
Argentina	1975-1976, 1982-1983, 1989-1991	1989
Brazil	1982-1983, 1987, 1990-1991, 1995	1987
Chile	1985	
Hong Kong (China, P.R.)		
Columbia	1985	1985
Costa Rica	1981	1981
Cyprus		
Indonesia	1978, 1983, 1986, 1997	1983, 1986
Jordan	1983, 1987-1989, 1992	1989, 1992
Korea	1980, 1997	
Malaysia	1986, 1997	1986
Malta	1992, 1997	1997
Mauritius	1979, 1981	
Mexico	1976, 1982, 1985, 1994-1995	1982, 1994
Panama		
Philippines	1983-1984, 1986, 1997	1983, 1986
Singapore	1975	1975
South Africa	1975, 1978, 1984-1986, 1996	1984
Thailand	1981, 1984, 1997	1981, 1997
Trinidad & Tobago	1985, 1988, 1993	1985
Tunisia	1993	
Turkey	1978-1980, 1994	1994
Uruguay	1982-1983	1982
Venezuela	1984, 1986, 1989, 1994-1996	1986, 1989, 1994

Footnotes:

¹ Milesi-Ferretti and Razin (1998) also explore the determinants of the costs of reversals. They consider a cross-section of current account reversal episodes in a sample of 86 low and middle-income developing economies for the period 1971-1992. They search for the determinants of reversals and find both domestic and external variables that seem to play a role in explaining their occurrence. They find that a less appreciated exchange rate, higher investment and more openness prior to the reversal make recovery faster and more pronounced.

² Milesi-Ferretti and Razin (2000) examine a cross section of currency crises and current account reversals in a broad set of 105 low and middle-income countries (for 1970-1996). They identify the determinants of reversals and currency crises in probit regression and also provide an analysis of the costs of each separately (in terms of output growth). They estimate a linear cross-sectional model with average output growth in the 3 years following the episode in question as the dependent variable and identify macroeconomic variables that determine the costs of such episodes (either a currency crisis or a current account reversal).

³ In the final step all variables are transformed by: $v_{it}^* = v_{it} - (1 - \theta_i) \overline{v_i}$ where $\theta_i = \sqrt{\frac{\sigma_{\epsilon}^2}{\sigma_{\epsilon}^2 + T_i \sigma_u^2}}$

where v_{it} denotes any of the aforementioned variables and v_i denotes a group mean and the variance components are the one obtained in first two steps. For exact details on the motivation and estimation procedure, see Hausman and Taylor (1981) and Greene (2001) respectively.

⁴ Assuming any of the other control variables is not exogenous does not change our empirical results

⁵ For a detailed survey of asymptotic consistency results and GMM estimation methods casting doubts on some of the results in this literature, see Arellano and Honoré (forthcoming) and Bond et al. (2001).

⁶ We use the Limdep software suite in all our estimations. We thank Professor William Greene for generously providing us with an update of the LIMDEP package and some of the statistical procedures used to estimate the models.

⁷ Our currency pressure measure of crises does not include episodes of defence involving sharp rises in interest rates. Data for market-determined interest rates are not available for much of the sample period in many of the countries in our dataset.

⁸ This approach differs from that of Kaminsky and Reinhart (1999), for example, who deal with episodes of hyperinflation by separating the nominal exchange rate depreciation observations for each country according to whether or not inflation in the previous 6 months was greater than 150 percent, and they calculate for each sub-sample separate standard deviation and mean estimates with which to define exchange rate crisis episodes.

⁹ Kaminsky and Reinhart (1999) use a three standard deviation cut-off. While the choice of cut-off point is arbitrary, Frankel and Rose (1996) suggest that the results are not very sensitive to the precise cut-off chosen in selecting crisis episodes. Our output equation estimates using "major" currency crises, evaluated with the 3-standard deviation threshold, are very similar to the benchmark crisis measure.

¹⁰ In contrast, Milesi-Ferretti and Razin (1998) examine changes in 3-year rolling averages of the current account. They also impose a condition that the maximum deficit after the reversal must be no larger than

the minimum deficit in the 3 years preceding the reversal. In our identification of reversal episodes we are interested in short-run to medium run phenomena and therefore we examine annual changes in the current account.

¹¹ Their work indicates that the exact (and admittedly ad-hoc) threshold used does not affect their results significantly. We present results to that effect as well.

¹² Bagnai and Manzoochi (1999) use a different statistical methodology to differentiate between permanent and transitory changes in current account trends and define permanent changes as reversal episodes. They find that GDP growth is not statistically significant in their regression for the determinants of positive reversals, giving some support to the Sudden Stop argument that reversal episodes are a leading causal factor in declining output growth.

¹³ The 'openness' variable is defined as the sum of imports and exports relative to GDP. Real exchange rate overvaluation is defined as deviations from a fitted trend in the real trade weighted exchange rate. The real trade-weighted exchange rate is the trade-weighted sum of the bilateral real exchange rates (defined in terms of CPI indices) against the U.S. dollar, the German mark, and the Japanese yen. The trade-weights are based on the average bilateral trade with the United States, the European Union, and Japan in 1980 and 1990.

¹⁴ The difference between the two estimates is not statistically significant.

¹⁵ Interestingly, both the coefficients on the lead and lag of capital account reversals are insignificantly different from zero.

¹⁶ A possible caveat is that while the drop in output is indeed sharp following a Sudden Stop, pre-crisis output growth could have been 'unsustainably' high as a result of the ex-ante capital inflows. This would suggest that our estimated effects are an upper bound. However, evidence presented by Gourinchas et al. (2001) suggests that domestic lending booms (which are partially correlated with capital inflows) do not increase output growth markedly.

¹⁷ The sample here is somewhat smaller than the one used in the results reported in Tables 4 and 5. The GMM estimator poses both data restrictions and restrictions on the models that could be estimated with our data.

¹⁸ We also restricted our sample for non-OECD countries with a population of more than one million.

¹⁹ The exact specification of the probit regression is taken from Glick and Hutchison, 2001. Details are available from the authors upon request.

²⁰ For a survey of sample selection correction methodologies see Blundell and Costa Días (2000) and for empirical examination of the IMR model see Heckman (1990).

²¹ See Rodrik (1999).

²² These predictions are derived from an output equation, not reported in the text for brevity, where two leads and lags of currency crises, capital flow reversals, and Sudden Stops are included. The control variables are the same as in the benchmark equations (e.g. Table 4, column 1).

Titles published in EPRU's Working Paper Series, beginning in November 1993:

2002

- 02-01 Wolfgang Eggert and Martin Kolmar: Information Sharing, Multiple Nash Equilibria, and Asymmetric Capital-Tax Competition
- 02-02 Carl-Johan Dalgaard and Claus Thustrup Kreiner: *Endogenous Growth: A Knife-Edge* or the Razor's Edge?
- 02-03 Marta Loi, Teresa Lloyd-Braga and Hans Jørgen Whitta-Jacobsen: Endogenous Business Cycles and Systematic Stabilization Policy
- 02-04 Wolfgang Eggert and Martin Kolmar: Contests with Size Effects.
- 02-05 Wolfgang Eggert and Laszlo Goerke, Fiscal Policy, Economic Integration and Unemployment.
- 02-06 Carl-Johan Dalgaard, Idle Capital and Long-Run Productivity.
- 02-07 Sylvester C.W. Eijffinger and Wolf Wagner, *Taxation if Capital is not Perfectly Mobile: Tax Competition versus Tax Exportation.*
- 02-08 Jesper Gregers Linaa, The Idiosyncrasy of Business Cycles Across EU Countries.
- 02-09 Michael M. Hutchison and Ilan Neuberger, *How Bad Are Twins? Output Costs of Currency and Banking Crises.*
- 02-10 Rasmus Fatum and Michael M. Hutchison, *ECB Foreign Exchange Intervention and the Euro: Institutional Framework, News and Intervention.*
- 02-11 Rasmus Fatum and Michael M. Hutchison, *Is Foreign Exchange Market Intervention an Alternative to Monetary Policy? Evidence from Japan.*
- 02-12 Michael M. Hutchison and Ilan Noy, Sudden Stops and the Mexican Wave: Currency Crises, Capital Flow Reversals and Output Loss in Emerging Markets.

- 01-01 Henrik Jensen: Optimal Degrees of Transparency in Monetary Policymaking
- 01-02 Thorvaldur Gylfason and Gylfi Zoega: Natural Resources and Economic Growth: *The Role of Investment*.
- 01-03 Wilhelm Kohler: International Fragmentation of Value-added Chains: How Does it Affect Domestic Factor Prices?

- 01-04 Mark A. Roberts and Eric O'N. Fisher: *Funded Pensions, Labor Market Participation, and Economic Growth.*
- 01-05 Søren Bo Nielsen, Pascalis Raimondos-Møller and Guttorm Schjelderup: Formula Apportionment and Transfer Pricing under Oligopolistic Competition.
- 01-06 Christian Keuschnigg and Søren Bo Nielsen: Public Policy for Venture Capital.
- 01-07 Søren Bo Nielsen, Pascalis Raimondos-Møller, and Guttorm Schjelderup: *Tax Spillovers under Separate Accounting and Formula Apportionment*.
- 01-08 Peter Birch Sørensen: International Tax Coordination: Regionalism versus Globalism.
- 01-09 Michael M. Hutchison: A Cure Worse Than the Desease? Currency Crises and the Output Costs of IMF-Supported Stabilization Programs.
- 01-10 Rasmus Lenz and Torben Tranæs: Job Search and Savings: Wealth Effects and Duration Dependence.
- 01-11 Dieter Bös: Bureaucrats and Public Procurement.
- 01-12 Kala Krishna and Cemile Yavas: *Wage Equality in a General Equilibrium Model with Indivisibilities.*
- 01-13 Mark A. Roberts: Funding the Transition from Pay-As-You-Go Pensions by Taxing Capital Gains on Land.
- 01-14 Wilfred J. Ethier: *Punishments and Dispute Settlement in Trade Agreements*.
- 01-15 Peter Christoffersen, Eric Ghysels, and Norman R. Swanson: Let's Get "Real" about Using Economic Data.
- 01-16 James E. Alt, David Dreyer Lassen, and David Skilling: *Fiscal Transparency, Gubernatorial Popularity, and the Scale of Government: Evidence from the States*

- 00-01 Andrew Hughes Hallett and Maria Demertzis: When Can An Independent Central Bank Offer Lower Inflation at No Cost? A Political Economy Analysis.
- 00-02 Torben M. Andersen: International Integration, Risk and the Welfare State.
- 00-03 F. Gulzin Ozkan, Anne Sibert and Alan Sutherland: *Monetary Union, Entry Conditions and Economic Reform.*
- 00-04 Michael M. Hutchison: European Banking Distress and EMU: Institutional and Macroeconomic Risks.

- 00-05 Bertil Holmlund: Labor Taxation in Search Equilibrium with Home Production.
- 00-06 Knud Jørgen Munk: Administrative Costs and the "Double Dividend".
- 00-07 Carl-Johan Dalgaard and Claus Thustrup Hansen: *Scale-Invariant Endogenous Growth*.
- 00-08 A. Lans Bovenberg and Ben J. Heijdra: *Environmental Abatement and Intergenerational Distribution*.
- 00-09 Jan Overgaard Olesen: A Simple Explanation of Stock Price Behaviour in the Long Run: Evidence for Denmark.
- 00-10 John E. Roemer, Rolf Aaberge, Ugo Colombino, Johan Fritzell, Stephen P. Jenkins, Ive Marx, Marianne Page, Evert Pommer, Javier Ruiz-Castillo, Maria Jesus San Segundo, Torben Tranæs, Gert G. Wagner, and Ignacio Zubiri: *To What Extent Do Fiscal Regimes Equalize Opportunities for Income Acquisition among Citizens?*
- 00-11 Christian Schultz and Tomas Sjöström: Local Public Goods, Debt and Migration.
- 00-12 Francesco Daveri: Is Growth an Information Technology Story in Europe Too?
- 00-13 Peter F. Christoffersen: Dating the Turning Points of Nordic Business Cycles.
- 00-14 Reuven Glick and Michael M. Hutchison: *Stopping "Hot Money" or Signalling Bad Policy? Capital Controls and the Onset of Currency Crises.*
- 00-15 Dan Anderberg and Carlo Perroni: *Renegotiation of Social Contracts by Majority Rule*.
- 00-16 Henrik Jacobsen Kleven: Optimum Taxation and the Allocation of Time.
- 00-17 Clemens Fuest, Bernd Huber and Søren Bo Nielsen: *Why Is the Corporate Tax Rate Lower than the Personal Tax Rate?*
- 00-18 Christian Keuschnigg and Søren Bo Nielsen: Tax Policy, Venture Capital, and Entrepreneurship.
- 00-19 Harry Huizinga and Søren Bo Nielsen: Withholding Taxes or Information Exchange: The Taxation of International Interest Flows.
- 00-20 David Dreyer Lassen: *Political Accountability and the Size of Government: Theory and Cross-Country Evidence.*

- 99-01 Henrik Jacobsen and Peter Birch Sørensen: Labour Tax Reform, *The Good Jobs and the Bad Jobs*.
- 99-02 Ignacio Ortuno-Ortin and Christian Schultz: *Divide the Dollar, A Model of Interregional Redistributive Politics.*

- 99-03 Pekka Ilmakunnas, Vesa Kanniainen, and Uki Lammi: Entrepreneurship, Economic Risks, and Risk-Insurance in the Welfare State.
- 99-04 Niels Thygesen: *Evolving Ambitions in Europe's Monetary Unification*.
- 99-05 Sajal Lahiri and Pascalis Raimondos-Møller: Lobbying by Ethnic Groups and Aid Allocation.
- 99-06 Roel M.W.J. Beetsma and Henrik Jensen: *Structural Convergence under Reversible and Irreversible Monetary Unification.*
- 99-07 Wolfgang Mayer and Pascalis Raimondos-Møller: The Politics of Foreign Aid.
- 99-08 Dermot Leahy and Catia Montagna: *Temporary Social Dumping, Union Legalisation and FDI: A Note on the Strategic Use of Standards.*
- 99-09 Rasmus Fatum and Michael M. Hutchison: Is Sterilized Foreign Exchange Intervention Effective After All? An Event Study Approach.
- 99-10 Christian Schultz and Tomas Sjöström: Public Debt, Property Values and Migration.
- 99-11 Roel M.W.J. Beetsma and Henrik Jensen: *Risk Sharing and Moral Hazard with a Stability Pact.*
- 99-12 Henrik Jacobsen Kleven, Wolfram F. Richter and Peter Birch Sørensen: *Optimal Taxation with Household Production.*
- 99-13 Huw Dixon, Claus Thustrup Hansen and Henrik Jacobsen Kleven: *Dual Labour Markets and Menu Costs: Explaining the Cyclicality of Productivity and Wage Differentials.*
- 99-14 Jim Malley and Hassan Molana: Fiscal Policy and the Composition of Private Consumption: Some Evidence from the U.S. and Canada.
- 99-15 Jonas Agell and Per Lundborg: Survey Evidence on Wage Rigidity and Unemployment: Sweden in the 1990s.
- 99-16 Satya P. Das: North-South Trade, Capital Accumulation and Personal Distribution of Wealth and Income.
- 99-17 Holger Bonin, Bernd Raffelhüschen and Jan Walliser: *Can Immigration Alleviate the Demographic Burden?*
- 99-18 Morten Hvidt and Søren Bo Nielsen: *Noncooperative vs. Minimum-Rate Commodity Taxation.*
- 99-19 Lisandro Abrego and Carlo Perroni: Investment Subsidies and Time-Consistent Environmental Policy.

- 99-20 Reuven Glick and Michael M. Hutchison: *Banking and Currency Crises: How Common Are Twins?*
- 99-21 Mark Gradstein and Moshe Justman: Public Schooling, Social Capital and Growth.
- 99-22 Jeremy S.S. Edwards and Alfons J. Weichenrieder: *Ownership Concentration and Share Valuation: Evidence from Germany.*
- 99-23 Henrik Jensen: Targeting Nominal Income Growth or Inflation?
- 99-24 Svend E. Hougaard Jensen and Thomas F. Rutherford: *Distributional Effects of Fiscal Consolidation*.

- 98-01 Pascalis Raimondos-Møller and Kimberley A. Scharf: *The Optimal Design of Transfer Pricing Rules: A Non-Cooperative Analysis.*
- 98-02 Michael Keen, Sajal Lahiri and Pascalis Raimondos-Møller: *When Is Policy Harmonisation Desirable?*
- 98-03 Clemens Fuest and Bernd Huber: *Tax Progression and Human Capital in Imperfect Labour Markets.*
- 98-04 Frank Hettich and Minna Selene Svane: *Environmental Policy in a Two Sector Endogenous Growth Model.*
- 98-05 Harry Huizinga and Søren Bo Nielsen: Is Coordination of Fiscal Deficits Necessary?
- 98-06 Claus Thustrup Hansen: A Note on Blanchard & Kiyotaki (1987).
- 98-07 Claus Thustrup Hansen: Long Run Impact of Increased Wage Pressure.
- 98-08 Erkki Koskela and Ronnie Schöb: Why Governments Should Tax Mobile Capital in the Presence of Unemployment.
- 98-09 Mark A. Roberts: Unfunded Social Security in the OLG Model with an Imperfectly Competitive Finance Market.
- 98-10 Peter Birch Sørensen: *Tax Policy, the Good Jobs and the Bad Jobs.*
- 98-11 Roel M.W.J. Beetsma and Henrik Jensen: Optimal Inflation Targets, "Conservative" Central Banks, and Linear Inflation Contracts: Comment.
- 98-12 Ole Risager: Random Walk or Mean Reversion: *The Danish Stock Market Since World War I.*

- 98-13 João Ejarque and Torben Tranæs: Skill-Neutral Shocks and Institutional Changes: Implications for Productivity Growth and Wage Dispersion.
- 98-14 Minna Selene Svane: Growth, Training Leave and Unemployment.
- 98-15 Svend E. Hougaard Jensen: *Nominal Stability, Real Convergence, and Fiscal Transfers in a Monetary Union.*
- 98-16 U. Michael Bergman and Michael Hutchison: *The Costs of EMU and Economic Convergence*.
- 98-17 Niels Thygesen: Fiscal Institutions in EMU and the Stability Pact.
- 98-18 Søren Bo Nielsen: A Simple Model of Commodity Taxation and Cross-Border Shopping.
- 98-19 Christian Schultz: Monetary Policy, Delegation and Polarization.
- 98-20 Knud Jørgen Munk: Should Governments Create Production Inefficiency?
- 98-21 Syed M. Ahsan and Panagiotis Tsigaris: *The Public Discount Rate and the Uncertain Budgetary Flows.*
- 98-22 Minna Selene Svane: *Emission Standards and Growth*.

- 97-01 Harry Huizinga and Søren Bo Nielsen: *The Political Economy of Capital Income and Profit Taxation in a Small Open Economy.*
- 97-02 Torsten Sløk and Jens Peter Sørensen: *How Small Shocks and Heterogeneous Expectations Can Create Swings in the Exchange Rate.*
- 97-03 Thórarinn G. Pétursson and Torsten Sløk: *Wage Formation in a Cointegrated VAR Model: A Demand and Supply Approach.*
- 97-04 Jeffrey H. Nilsen: Borrowed Reserves, Fed Funds Rate Targets, and the Term Structure.
- 97-05 Carlo Perroni and Kimberley A. Scharf: *Tiebout with Politics: Capital Tax Competition and Constitutional Choices.*
- 97-06 Sajal Lahiri, Pascalis Raimondos-Møller, Kar-yiu Wong, and Alan D. Woodland: *Optimal Income Transfers and Tariffs.*
- 97-07 Claus Thustrup Hansen and Hans Jørgen Jacobsen: *Rebalancing Unemployment Benefits in a Unionized Labour Market.*
- 97-08 Sören Blomquist and Vidar Christiansen: Price Subsidies versus Public Provision.

- 97-09 Amrita Dhillon, Carlo Perroni and Kimberley A. Scharf: *Implementing Tax Coordination*.
- 97-10 Peter Birch Sørensen: Optimal Tax Progressivity in Imperfect Labour Markets.
- 97-11 Syed M. Ahsan and Peter Tsigaris: *The Design of a Consumption Tax under Capital Risk.*
- 97-12 Claus Thustrup Hansen and Søren Kyhl: *Pay-per-view Television: Consequences of a Ban.*
- 97-13 Harry Huizinga and Søren Bo Nielsen: *The Taxation of Interest in Europe: A Minimum Withholding Tax?*
- 97-14 Harry Huizinga and Søren Bo Nielsen: A Welfare Comparison of International Tax Regimes with Cross-Ownership of Firms.
- 97-15 Pascalis Raimondos-Møller and Alan D. Woodland: *Tariff Strategies and Small Open Economies*.
- 97-16 Ritva Tarkiainen and Matti Tuomala: On Optimal Income Taxation with Heterogenous Work Preferences.
- 97-17 Minna Selene Svane: Optimal Taxation in a Two Sector Model of Endogenous Growth.
- 97-18 Frank Hettich: Growth Effects of a Revenue Neutral Environmental Tax Reform.
- 97-19 Erling Steigum, Jr.: Fiscal Deficits, Asset Prices and Intergenerational Distribution in an Open Unionized Economy.
- 97-20 Rod Falvey and Geoff Reed: Rules of Origin as Commercial Policy Instruments.
- 97-21 U. Michael Bergman, Michael M. Hutchison and Yin-Wong Cheung: *Should the Nordic Countries Join A European Monetary Union? An Empirical Analysis.*
- 97-22 Kenneth M. Kletzer: *Macroeconomic Stabilization with a Common Currency: Does European Monetary Unification Create a Need for Fiscal Insurance or Federalism?*
- 97-23 Martin Richardson: Trade Policy and Access to Retail Distribution.
- 97-24 Sugata Marjit and Hamid Beladi: Protection, Underemployment and Welfare.
- 97-25 Bernd Huber: Tax Competition and Tax Coordination in an Optimum Income Tax Model.
- 97-26 Clemens Fuest and Bernd Huber: *Tax Coordination and Unemployment*.
- 97-27 Assaf Razin, Efraim Sadka, and Chi-Wa Yuen: *Quantitative Implications of the Home Bias: Foreign Underinvestment, Domestic Oversaving, and Corrective Taxation.*

97-28 Mark A. Roberts, Karsten Stæhr, and Torben Tranæs: *Two-Stage Bargaining with Coverage Extension in a Dual Labour Market*.

- 96-01 Torben Tranæs: A Simple Model of Raiding Opportunities and Unemployment.
- 96-02 Kala Krishna and Ling Hui Tan: *Transferable Licenses vs. Nontransferable Licenses: What is the Difference?*
- 96-03 Jiandong Ju and Kala Krishna: Market Access and Welfare Effects of Free Trade. Areas without Rules of Origin.
- 96-04 Anders Sørensen: Growth Enhancing Policies in a Small Open Economy.
- 96-05 Anders Sørensen: Industrialization and Factor Accumulation.
- 96-06 Christian Schultz: Announcements and Credibility of Monetary Policy.
- 96-07 Christian Schultz: Political Competition and Polarization.
- 96-08 Ole Risager and William G. Tyler: *Macroeconomic Policy and Exchange Rate Policy Management in a Small Dependent Economy: Estimating the Effects of Currency Devaluation in Jordan.*
- 96-09 Neil Rankin: How Does Uncertainty About Future Fiscal Policy Affect Current Macroeconomic Variables?
- 96-10 U. Michael Bergman and Michael M. Hutchison: *The 'German View', Fiscal Consolidation and Consumption Booms: Empirical Evidence from Denmark.*
- 96-11 Eric Hansen and Michael M. Hutchison: *Exchange Rates, Non-traded Goods and the Terms-of-Trade: An Empirical Application for New Zealand.*
- 96-12 Michael M. Hutchison and Carl E. Walsh: *Central Bank Institutional Design and the Output Cost of Disinflation: Did the 1989 New Zealand Reserve Bank Act Affect the Inflation-Output Tradeoff?*
- 96-13 Rasmus Fatum and Michael M. Hutchison: Is Intervention a Signal of Future Monetary Policy? Evidence from the Federal Funds Futures Market.
- 96-14 Tryggvi Thor Herbertsson and Anders Sørensen: Policy Rules for Exploitation of Renewable Resources: A Macroeconomic Perspective.
- 96-15 Anders Sørensen: International Welfare Effects from Country-Specific R&D Subsidies.
- 96-16 Andreas Haufler and Søren Bo Nielsen: *Dynamic Effects of an Anticipated Switch from Destination- to Origin-Based Commodity Taxation.*

- 96-17 Harry Huizinga and Søren Bo Nielsen: *The Coordination of Capital Income and Profit Taxation with Cross-Ownership of Firms.*
- 96-18 Svend Erik Hougaard Jensen: Wage Rigidity, Monetary Integration and Fiscal Stabilisation in Europe.
- 96-19 Ole Risager and Jan Rose Sørensen: Job Security Policies and Trade Union Behaviour in an Open Economy.
- 96-20 Slobodan Djajic, Sajal Lahiri and Pascalis Raimondos-Møller: Transfer and the Intertemporal Terms of Trade.
- 96-21 Slobodan Djajic, Sajal Lahiri and Pascalis Raimondos-Møller: Logic of Aid in an Intertemporal Setting.
- 96-22 Svend E. Hougaard Jensen and Bernd Raffelhüschen: *Public Debt, Welfare Reforms, and Intergenerational Distribution of Tax Burdens in Denmark.*

- 95-01 Vesa Kanniainen and Jan Södersten: On Financial Adjustment and Investment Booms: Lessons from Tax Reforms.
- 95-02 Søren Bo Nielsen: Withholding Taxes and Country-Specific Shocks.
- 95-03 Vesa Kanniainen and Rune Stenbacka: *Towards a Theory of Socially Valuable Imitation* with Implications for Technology Policy.
- 95-04 Bent E. Sørensen, Pierfederico Asdrubali, and Oved Yosha: *Channels of Interstate Risksharing: US 1963-1990.*
- 95-05 Peter Birch Sørensen: Changing Views of the Corporate Income Tax.
- 95-06 Robin Boadway: The Role of Second-Best Theory in Public Economics.
- 95-07 Sjak Smulders: Environmental Policy and Sustainable Economic Growth an endogenous growth perspective.
- 95-08 Bernd Genser: Patterns of Tax Arbitrage and Decentralized Tax Autonomy.
- 95-09 Harry Huizinga and Søren Bo Nielsen: Capital Income and Profits Taxation with Foreign Ownership of Firms.
- 95-10 Ben Lockwood: Commodity Tax Harmonisation with Public Goods an Alternative Perspective.
- 95-11 Saqib Jafarey, Yannis Kaskarelis, and Apostolis Philippopoulos: *Private Investment and Endogenous Fiscal Policy. Theory and Evidence from UK and USA.*

- 95-12 Svend Erik Hougaard Jensen: *Debt Reduction, Wage Formation and Intergenerational Welfare.*
- 95-13 Sajal Lahiri and Pascalis Raimondos: *Public Good Provision and the Welfare Effects of Indirect Tax Harmonisation*.
- 95-14 Ruud A. de Mooij and A. Lans Bovenberg: *Environmental Taxes, International Capital Mobility and Inefficient Tax Systems: Tax Burden vs. Tax Shifting.*
- 95-15 David F. Bradford: Consumption Taxes: Some Fundamental Transition Issues.
- 95-16 Ole Risager: On the Effects of Trade Policy Reform: The Case of Jordan.
- 95-17 Niels Thygesen: The Prospects for EMU by 1999 and Reflections on Arrangements for the Outsiders.
- 95-18 Christian Keuschnigg and Søren Bo Nielsen: Housing Markets and Vacant Land.
- 95-19 Hans Fehr: Welfare Effects of Investment Incentive Policies: A Quantitative Assessment.
- 95-20 Ben Lockwood, Torsten Sløk, and Torben Tranæs: *Progressive Taxation and Wage Setting: Some Evidence for Denmark.*
- 95-21 Claus Thustrup Hansen, Lars Haagen Pedersen, and Torsten Sløk: *Progressive Taxation, Wages and Activity in a Small Open Economy.*
- 95-22 Svend Erik Hougaard Jensen and Bernd Raffelhüschen: *Intertemporal Aspects of Fiscal Policy in Denmark.*

- 94-01 Niels Thygesen: Reinforcing Stage Two in the EMU Process.
- 94-02 Kåre P. Hagen and Vesa Kanniainen: Optimal Taxation of Intangible Capital.
- 94-03 Ed W.M.T. Westerhout: The Economic and Welfare Effects of Taxing Foreign Assets.
- 94-04 Slobodan Djajic: Illegal Immigration and Resource Allocation.
- 94-05 Sajal Lahiri and Pascalis Raimondos: Is There Anything Wrong with Tied-Aid?
- 94-06 Ben Lockwood, Apostolis Philippopoulos, and Andy Snell: *Fiscal Policy, Public Debt Stabilzation and Politics: Theory and evidence from the US and UK.*
- 94-07 Partha Sen: Welfare-Improving Debt Policy under Monopolistic Competition.
- 94-08 Sajal Lahiri and Pascalis Raimondos: *Tying of Aid to Trade Policy Reform and Welfare*.
- 94-09 Mark Gradstein and Moshe Justman: *Public Choice of an Education System and its Implications for Growth and Income Distribution.*

- 94-10 Peter Birch Sørensen, Lars Haagen Pedersen, and Søren Bo Nielsen: *Taxation, Pollution, Unemployment and Growth: Could there be a "Triple Dividend" from a Green Tax Reform?*
- 94-11 Peter Birch Sørensen and Søren Bo Nielsen: On the Optimality of the Nordic System of Dual Income Taxation.
- 94-12 Sajal Lahiri and Pascalis Raimondos: Competition for Aid and Trade Policy.
- 94-13 Niels Kleis Frederiksen, Peter Reinhard Hansen, Henrik Jacobsen, and Peter Birch Sørensen: *Comsumer Services, Employment and the Informal Economy.*
- 94-14 Yoshiyasu Ono: Market Segmentation and Effective Demand Shortage in a World with Dynamic Optimization.

- 93-01 Svend Erik Hougaard Jensen, Søren Bo Nielsen, Lars Haagen Petersen and Peter Birch Sørensen: Tax Reform, Welfare, and Intergenerational Redistribution - An Intertemporal Simulation Approach.
- 93-02 Bernd Genser, Andreas Haufler and Peter Birch Sørensen: Indirect Taxation in an Integrated Europe. Is there a Way of Avoiding Trade Distortions Without Sacrificing National Tax Autonomy?
- 93-03 Svend Erik Hougaard Jensen and Lars Grue Jensen: *Debt, Deficits and Transition to EMU: A Small Country Analysis.*
- 93-04 Torsten Persson and Guido Tabellini: *Federal Fiscal Constitutions. Part I: Risk Sharing and Moral Hazard.*
- 93-05 Martin Paldam: The Political Economy of Stopping High Inflation.
- 93-06 Roger H. Gordon and Jeffrey K. Mackie-Mason: Why is There Corporate Taxation in a Small Open Economy? The Role of Transfer Pricing and Income Shifting.
- 93-07 Peter Birch Sørensen: From the Global Income Tax To the Dual Income Tax: Recent Tax Reforms in The Nordic Countries.
- 93-08 Wilhelm Kohler: *Strategic Trade Policy and Integration*.
- 93-09 F. Gulcin Ozkan and Alan Sutherland: A Model of the ERM Crisis.